Author’s response to reviews

Title: Factors associated with dental visit and barriers to utilisation of oral health care services in a sample of antenatal mothers in Hospital Universiti Sains Malaysia

Authors:

Norkhafizah Saddki (fizah@kb.usm.my)
Azizah Yusoff (drazizah@kck.usm.my)
Yew L Hwang (yewlang83@yahoo.com)

Version: 4 Date: 11 February 2010

Author’s response to reviews:

Dear Sir/Madam,

The point-by-point responses to the formatting changes suggested are as follows:

Major revisions:

1. Title: Changes were made at the submission system. The title now does match.

2. Author list: Changes were made accordingly in the manuscript. The authors names now do match.

3. Abstract: Changes were made at the submission system. The abstract now does match.

Minor revisions:

1. References:

Reference 27 – link/URL for this reference provided

Reference 37 – this journal is published exclusively online, there is no page number

2. Tables: The authors are not sure and understand how to go about in doing it. Please provide assistance.

3. Typography: Checked

Thank you.

Regards,
Dr. Norkhafizah Saddki